Author's response to reviews

Title: Trancutanous vagus nerve stimulation for the treatment of depression: a study protocol for a double blinded randomized clinical trails

Authors:

Peijing Rong (rongpj@hotmail.com)
Jiliang Fang (fangmgh@gmail.com)
Jian Kong (kongji@nmr.mgh.harvard.edu)
Liping Wang (dwlp@vip.sina.com)
Hong Meng (menghong2000@163.com)
Jun Liu (tracyliu518@hotmail.com)
Yingge Ma (magican-111@163.com)
Hui Ben (bh9997@163.com)
Liang Li (liiang1160@139.com)
Rupeng Liu (acupeng@hotmail.com)
Xia Li (lixiafei1988@126.com)
Bing Zhu (zhubing@mail.cintcm.ac.cn)

Version: 4 Date: 12 November 2012

Author's response to reviews: see over
Nov. 12, 2012

Dear editors:

After receiving your kind advice, we do some revise on our manuscript. We put the details of the subject who provide the ear image on figure 2. This subject confirm that he permit that he voluntarily and without compensation authorize the photograph of his ear to be used by BMC.

Please go through our manuscript entitled “Trancutaneousvagus nerve stimulation for the treatment of depression: a study protocol for a double blinded randomized clinical trial” continuously.

This clinical trial is funded by National Basic Research Program of China (973 program, no. 2012CB 518503), National twelfth five-year science and technology support program of China (2012BAF14B10) and Beijing Natural Science Foundation of China (no. 711117). There is no conflict of interest to claim for all authors.

Thank you for your consideration.

Best Regards

Peijing Rong  M.D., Ph.D.
Institute of Acupuncture and Moxibustion,
China Academy of Chinese Medical Sciences
rongpj@hotmail.com